Author’s response to reviews

Title: Association between Serum Uric Acid and Metabolic Syndrome Components in Prepubertal Obese Children (Tanner Stage I) from Nuevo León, Mexico - A Preliminary Study

Authors:

Elizabeth Solis-Perez (elizabeth.solis@uanl.mx)

Mario Medina (mario.gonzalez@iiiepe.edu.mx)

Manuel Lomeli (lolm27@hotmail.com)

Verónica González (veronica.tijerinag@uanl.mx)

Jesús Pérez (zacvilla@yahoo.com.mx)

Fernando González (dfernandolavalle@hotmail.com)

Victorine Imrhan (VImrhan@mail.twu.edu)

Shanil Juma (SJuma@twu.edu)

Parakat Vijayagopal (PVijayagopal@mail.twu.edu)

Kittipong Boonme (kboonme@mail.twu.edu)

Chandan Prasad (cprasad@twu.edu;cprasad41@gmail.com)

Version: 3 Date: 15 Jun 2017

Author’s response to reviews:

June 15, 2017

Editors BMC-Obesity

Thank you very much for making last minute suggestions for improving the manuscript. We have made all recommended changes as follows:
(1) “Consent for publication” declaration section has been added

(2) Author’s contribution section has been edited to include initials only

(3) Figure legends have been placed after references

(4) All 4 illustrations of Figure 2 has been re-formatted to fit as a single panel.

Sincerely,

Chandan Prasad, PhD